Author's response to reviews

Title: alpha-Mangostin extracted from the pericarp of the mangosteen (Garcinia mangostana Linn) reduces tumor growth and lymph node metastasis in an immunocompetent xenograft model of metastatic mammary cancer carrying a p53 mutation

Authors:

Masa-Aki Shibata (masaaki.shibata@ohsu.ac.jp)
Munekazu Iinuma (iinuma@gifu-pu.ac.jp)
Junji Morimoto (eac001@art.osaka-med.ac.jp)
Hitomi Kurose (an1021@art.osaka-med.ac.jp)
Kanako Akamatsu (k.akamatsu@pharm.kyoto-u.ac.jp)
Yasushi Okuno (okuno@pharm.kyoto-u.ac.jp)
Yukihiro Akao (yakao@giib.or.jp)
Yoshinori Otsuki (an1001@art.osaka-med.ac.jp)

Version: 4 Date: 8 May 2011

Author's response to reviews: see over
Journal: BMC Medicine, MS: 1255334126508757, Formatting changes

(This article has been accepted.)

May 8, 2011
Dear Sir and madam,

Since the manuscript did not completely match to the latest style of BMC Medicine, the manuscript has been arranged.

Thank you very much.
Sincerely yours,

Masa-Aki Shibata
Osaka Health Science University